Author's response to reviews

Title: Duloxetine in treatment of refractory chronic tennis elbow: two case reports

Authors:

Zaid A Wani (zaidwani@gmail.com)
Shabir A Dhar (shabirdhar@yahoo.co.in)
Muhammad F Bhat (mfbutt72@yahoo.com)
Yasir H Rather (yasirhrather@yahoo.com)
Shano Shiekh (zaidwani@yahoo.com)

Version: 4 Date: 15 July 2008

Author's response to reviews:

Dear Sir

Thanks for letting us to revise our manuscript titled Duloxetine in treatment of refractory chronic tennis elbow: two case reports

As per the reviewers comments the following changes have been made

1. Musculoskeletal added in the first paragraph of introduction.

2. Introduction 3 rd paragraph has been re written to include the comments of the reviewers .

3. Case histories – unnecessary content removed and cases made concise.

4. Abstract has been tailored and case histories condensed .

5. Conclusion rewritten to include comments about duloxetine as an alternative treatment .

6. References citing Use of duloxetine in neuropathy and fibromyalgia used rather than that of venlafaxine which has been removed .

7. Reviewers comments about the case histories about duration of duloxetine and VAS added.

8. Reference of Venlafaxine removed.

9. Discussion and conclusion rewritten as per the reviewers comments.

10. Language corrections made where ever required .

Dr zaid wani